Author's response to reviews

Title: Associations of selection criteria and socio-demographic factors in medical students with intended rural site of practice following graduation

Authors:

Ian B Puddey (ian.puddey@uwa.edu.au)
Annette Mercer (annette.mercer@uwa.edu.au)
Denese E Playford (denese.playford@rcswa.edu.au)
Sue Pougnault (sue.pougnault@uwa.edu.au)
Geoffrey J Riley (geoff.riley@rcswa.edu.au)

Version: 2
Date: 29 May 2014

Author's response to reviews:

The following statement has been included at lines 241 and 242 "all MSOD participants provided informed written consent."

Line numbers have been added sequentially throughout the manuscript as requested.